
\documentclass[11pt]{article}
\usepackage{amsfonts}
\addtolength{\textwidth}{\oddsidemargin}  %i.e., left margin
\addtolength{\textwidth}{0.75in}
\addtolength{\textheight}{1in}
\setlength{\oddsidemargin}{0in}
\setlength{\topmargin}{0in}
\setlength{\headheight}{0in}
\setlength{\headsep}{0in}

\begin{document}
\begin{center}
Intro to Mathematical Reasoning (Math 300) -- 
Homework 8\footnote{Version 3/9/04}
\end{center}

\begin{enumerate}
\item Exercise 5.2.2

\item Prove Theorem  5.2.3

\item Problem 5.2.4

\item
Prove Theorem 5.2.9

\item End of chapter 5, problem 1
\item End of chapter 5 problem 16. (Note: there is a typographical
error in the statement of the problem: in line 2
it should say $\mathcal{F}:\mathcal{B} \longrightarrow 
\mathcal{A}$.
\end{enumerate}


\end{document}










